1 ' 19z KusProtiEloCourse Overviewuung Users andTheir Tasksuples and Guideug With Deun StylesuUI Design Elemul Design GuideuUI Devent Toolsue Designuct Presentations andted TopicsuCase Sunt Develts inus ' 19z KusProtiEloChapter OverviewIterative Design and PrototypinguuuIterative DuSoftware ring Lifeeuuuping Tecuts and sut Concepts andTermsuter Summary ' 19z KusProtiEloMotivationuinterface inflcle consiuthe user interfpressions of users t the ovuups, scenarios, prototypuung errors late in the t process is emore eufor many software systems, modifications based ontisfied users are a very large prall costs ' 19z KusProtiElout ofthe user interface fits into the overall suof user interface prototyuototypes for ' 19z KusProtiEloPrototypesusimulate the structure, functionalityanother sysusent a model of the aption, service, or prodbe buumay or may not have any real functionucan be either paper based or computer ba[Mo] ' 19z KusProtiEloPaper-based Prototypesuulow fidelityuue sketchuionality so that users can actuallyinteract [Mo]

2 ' 19z KusProtiEloComputer-based Puhigher fidelity uof functiucan offer vlook like[Mo] ' 19z KusProtiEloWhy Prototype?upart of tuuucost of corras the software life c[Mo] ' 19z KusProtiElo0Prototyping Techniquesulow-fidelity uhigh-fidelity [Mo] ' 19z KusProtiElo1Low-fidelity Prototypesulow-fidelity up, rapid versions otemud functionality and/or interactivityuct concepts, designs, alternatives, and screen larather than model user interaction with a systemve.g. storyboard ions, ept protounstrate the gl �feel and lookvtheir purpose is not to show in detail how the ation ouare often used early in the design vto show general ptual approaches t investing tmuch time or effort[Mo] ' 19z KusProtiElo2High-fidelity Prototypesuhigh-fidelity ufully invusers can enter data into entry , respond to mes, sels to open windows, and interact with the UIusent the core functionality of the puly builch as Smalltalk or Visual Bavcan ste much of the onality of the final sutrade ovnot as quick and easy to create as low-fidelity vully represent the UI to be implemented in the pvcan be almost identical in appearance to the actual produ[Mo] ' 19z KusProtiElo3Comparisoneity Limited error Evaluate different Addresses scrLimited usefulness after reqUseful for identifying market requirementsProof of conceptty High degree of More expensive Time consuming tInefficient for proof of concept Not effective for r gatheringUseful for exploration & test Look and feel of final productServes as a Marketing and [Mo]

3 ' 19z KusProtiElo4Fidelity Requirementsurecent sprototype fidelity u3 levels of prototypes:vpaper - low fidelivn shots - medium fidelvtive Visual c - high fidel[Mo] ' 19z KusProtiElo5Case Study (cont.)u30 university stts al librarysearch tasks utotal of 99 uity problems were uno significin the specifusers ubility Evnand Prototype Fidelity: Users and Usability Pros of the Human Factors and Ergonty, 42nd Annual Meeting, 13[Mo] ' 19z KusProtiElo6Low-fidelity Prototypinguidentify key markuprovide a very high-level vand servicuuuof platform, uuget custuelicit to select[Mo] ' 19z KusProtiElo7High-fidelity Prototypingucreate a living specification for uion with customers to sce, or applicatuprior to the code being fully deveutest UI issues prior to committiung, instru[Mo] ' 19z KusProtiElo8Software Prototypesuactually work to sunot an idea or duquickly uway - thrown away or discarded imtely afterumental - separate conts, addeuy - may eventually evolve into the final systemumay serve many diffut user reactions, serve auan iterative procudes modn and ev[Mo] ' 19z KusProtiElo9Levels of Prototypingufull prototyuuu

4 ' 19z KusProtiElo0Full Prototypeucontains cunal sysue.g. trial system with a limited number of simultaneumay be n[Mo] ' 19z KusProtiElo1Horizontal Prototypeuof a syudo not provide full functioue.g. users can execute all non and snds, but without retrieving any real inn as auuall of the features p[Mo] ' 19z KusProtiElo2Vertical Prototypeufunctionality, but only for a restricted partof the syuity in one or two modules, but not entiresystemue.g. in an e flight inrs canaccess a se with some real dtion providers, but not the entire datavin other words, they can play with a part of the uwith full[Mo] ' 19z KusProtiElo3ufunctionality er ofuvery uuser follows a previously usmall, tly uunctionality [Mo] ' 19z KusProtiElo4Diagram LevelsLrototynHzolproe

Valproe

S[Mo] ' 19z KusProtiElo5Chauffeured Prototypinguinvolves � the syully a member of the dement teamuthe system may not yet be cuser to test ituit is neves important to establish whether aence of actions is correct[Mo]

5 ' 19z KusProtiElo6Wizard of Ozuthe syuuser is acting as the systuusually cuto gain an unding of the user�s exps[Mo] ' 19z KusProtiElo7Testing of Prototypesuurve typical users attempting to execute typical taskson a prototype systemunote number of errors and where they occur, conns, and couuted toward testing the prototype any preished pce goavrformance in less than 30 min[Mo] ' 19z KusProtiElo8Testing of Prototypes (couutwo or more UI design (prototype) alternatives with thesame funlity are directly computhe one that leadsults is selected for thefinal p[Mo] ' 19z KusProtiElo9Benefits of Prototypingural part of the iterative design pugn valius issues not usually considered until deventudes a meact- or appl-fic questions that cannot be and by genrch or existing guiuts valuable user fek to be obtained early in then process[Mo] ' 19z KusProtiElo0Benefits of Prototyping (cont.)uitative and quantitative human pee data canbe collected within the cecific apudes a relatively cheap and easy wearly in the design cycleuts iterative etion and ng unding ofesign to the final puves the quality and completeness oal spenutially reduces the total dev[Mo] ' 19z KusProtiElo1ute analyvuate undeng of the ing problvthe lack of a thorough nding of the applon, service, product being duthe prototype may look like a completed system-ue expectavstic tations with respect to actual product performanceuring reavtations and cts that apply to the real product may be ignored within the prototyping s-[Mo]

6 ' 19z KusProtiElo2Drawbacks (contuusers that are never satisfiedvusers can ask for things that are beyond the e of the ung the prototype as an exercvopers may develop the wrong thivat great effort and euthe trap of over-design or unv�just one more feature ...�v�this is just the prototype, we�ll fix it when we develop the produc[Mo] ' 19z KusProtiElo5Important Concepts and Termsuve inteutic evuuutive wuxtual iuumental uexpert ufocus uve evanutic evuntal protouhuman factors uuuurapid protguuative uyuuuuser interface uuser obuuser requual protou ' 19z KusProtiElo6Chapter Summaryuuser interface design is allhe sysus in early stages of uties tas possiuuntered design and evan ' 19z KusProtiElo7